Author's response to reviews

Title: Dual method use for protection of pregnancy and disease prevention among HIV-infected women in South East Nigeria

Authors:

LUCKY O LAWANI (lawkins2000@yahoo.com)
CHUKWUEMEKA A IYOKE (caiyoke@yahoo.co.uk)
AZUBUIKE K ONYEBUCHI (kanayo009@yahoo.com)

Version: 5 Date: 19 February 2014

Author's response to reviews: see over
Response to reviewer’s comment

1. The title of the article has been rephrased as “dual method use for protection of pregnancy and disease prevention among HIV-infected women in South East Nigeria” to reflect the focus of the article and the national PMTCT recommendation which is the use of dual method as protection against STI’s/HIV and unplanned pregnancy.

2. The article has been edited to correct the grammatical errors.

3. The updated websites for the references are:

   http://www.who.int/topics/sexually_transmitted_infections/en/